ORIGINAL PAPER



# Population and development redux

Javier A. Birchenall<sup>1</sup>

Received: 30 September 2014 / Accepted: 18 September 2015 © Springer-Verlag Berlin Heidelberg 2015

**Abstract** The effects of population growth on long-term economic development are obviously important. This paper introduces new predictions from a general Malthus-Boserup model of population growth and ideas-based technological change. It also tests these predictions using numerous data sources, empirical specifications, and sample periods. Time series tests reveal that the empirical associations that hold true in the modern era are completely reversed in pre-modern samples. Inferences drawn from the pre-modern population growth of geographically isolated populations are also reversed when relevant controls are taken into account. While there is a clear break with Malthusian theory, in general, and especially outside of the modern era, there is no unequivocal evidence supporting Boserupian views. An alternative model consistent with transitional demographic patterns is briefly discussed.

Keywords Population growth · Technological change · Malthus and Boserup

JEL Classification J10 · O40

Responsible editor: Alessandro Cigno

Javier A. Birchenall Javier.Birchenall@ucsb.edu

<sup>1</sup> Department of Economics, University of California, 2127 North Hall, Santa Barbara, CA 93106, USA

#### 1 Introduction

The relationship between population growth and economic development is contentious. Social scientists are broadly divided between Malthusians, who see population growth as a barrier to economic development, and Boserupians, who see population growth as a driver of technological change. The nature of this relationship has important implications for a range of issues in economics. Global environmental sustainability, sub-Saharan Africa's demographic momentum, the emergence of below-replacement fertility in developed countries, and demographic policies such as China's one-child policy are among the many current issues confronted by this perennial debate.

Since the earlier empirical assessments of this relationship in the 1960s and 1970s (see Kuznets (1960) among many others), there has been a general understanding that the modern era has not met the dismal predictions of the Malthusian model: population growth during the modern demographic transition did not bring global poverty. Although living standards have diverged in ways consistent with lingering Malthusian effects, assessments of the relationship between population growth and economic growth during the modern era have been generally inconclusive; see, e.g., Kelley (1988) and Pritchett (1996). As succinctly stated by Headey and Hodge (2009, p. 221) in their recent meta-study of the macro-literature, "a stylized fact in the macroeconomic literature on population growth is the absence of a robust effect of total population growth on economic growth."

Modern technological change has been able to accommodate a much larger global population size than Malthusian theory predicted. With the demise of Malthusian ideas, a disconnection between population growth and technological change gained acceptance. This disconnection was so entrenched that population growth and technological change were made into exogenous and independent variables by neoclassical growth theorists. Population growth was not banished from growth theory but, for the most part, studies of the relationship between population and economic growth went into hibernation. (There are, of course, many important exceptions. Assuming an exogenous population growth turned out to be critical, for example, for the stability of economic growth models with exhaustible resources; see Cigno (1981).)

Endogenous growth theories awakened the debate between Malthusians and Boserupians. Kremer (1993), in particular, added several new elements to the debate. He proposed a synthesis of Malthusian and Boserupian views. In this synthesis, as well as in more general models of endogenous growth, technological change is not passive; as in Boserup (1981), it actively responds to population size or the growth rate of population, depending on specific functional form assumptions. Kremer (1993) also provided novel and relevant empirical evidence for assessing these views of population growth. Based on long time series population data and a *natural experiment*, the geographic isolation of populations following the last glaciation, Kremer (1993) argued that the long-term history of population growth, a history dating

back to 1 million years ago (1 MYA), is consistent with a view whereby population growth spurs technological change.<sup>1</sup>

This paper presents some new evidence on the long-term economic effects of population growth. I introduce several new population growth predictions from a slightly more general model than Kremer (1993) and Klasen and Nestmann (2006). I assess these predictions and reassess some of the current findings using alternative demographic data. The findings largely differ from those in the literature in all possible dimensions. The model, for example, predicts population *growth effects* from exogenous factors such as arable land and biogeographic endowments. This testable prediction is a novel way to set Boserupian and Mathusian views apart. Malthusian theory, for instance, predicts *level effects* but no growth effects from exogenous variables; see, e.g., Ashraf and Galor (2011). I find that biogeographic variables and arable land are strongly associated with population growth. I also find, however, that the sign of this association is the opposite in modern and pre-modern data.

The model also predicts a positive association between population growth and population size. In samples that end after the 1970s, population growth and size are strongly positively associated, even in data not previously examined. But tests of this prediction are also completely reversed when recent observations are excluded. If recent observations are omitted, data as recent as those based on post-1900 observations, the findings display a "wrong" (e.g., negative) sign.

Finally, the model predicts a positive effect of initial population size on the growth rate of population in geographically isolated populations. The melting of the ice caps provided a *natural experiment* upon which this prediction can be tested; see, e.g., Kremer (1993). The last glaciation isolated the Old and the New Worlds, as well as Australia. If population densities were the same some 12 thousand years ago (12 KYA), one should expect that a smaller initial population size (a consequence of its smaller area, since densities are assumed equal) would generate a technological disadvantage for small areas circa 1500 AD. This comparison, however, disregards differences in factor endowments and biogeography. This omission is problematic because the importance assigned to initial population size could simply be due to geographic differences, which, according to Diamond (1997), favored an early origin of agriculture and an easier diffusion of post-agricultural technologies such as metallurgy and weaponry in the Old World. Once systematic differences are taken into account, I find no effect of initial population size on population growth rates.

<sup>&</sup>lt;sup>1</sup>Kuznets (1960) and Simon (1977) emphasized, but did not quantify, positive population externalities such as those advocated by endogenous growth theorists. Johnson (2000) and Jones (2005) also discussed population externalities in the production of knowledge in a long-term perspective. Pryor and Maurer (1982) and Lee (1988) are earlier syntheses of Malthus and Boserup with predictions that are similar to those of Kremer (1993). Curiously, the (positive) relationship between population growth and population size was first studied in the 1960s in the context of fatalistic "doomsday" models; see, e.g., von Foerster et al. (1960) and Umpleby (1987).

The main lesson one can draw from the present exercises is that pre-modern and modern demographic regimes are so different that the implied long-term relationship between population growth and technological change cannot be satisfactorily interpreted using a Boserupian perspective. In general, the empirical associations that hold true in the modern era are contradicted in pre-modern samples. For example, in pre-modern samples, population exhibits *mean reversion*, as a stationary Malthusian equilibrium would predict.<sup>2</sup> There is no support for Malthusian predictions in the modern era; in modern samples, there is no mean reversion in population. Likewise, population size and biogeographic endowments influence population growth in *opposite ways* in modern and pre-modern samples.

The sharp contrast between pre-modern and modern demographic regimes runs against traditional unified growth models and ideas-based models of technological change.<sup>3</sup> In these models, long-run economic growth depends on population dynamics; see, e.g., Galor and Weil (2000), Jones (2005), Kremer (1993), Kortum (1997), and Segerstrom (1998).<sup>4</sup> The absence of Boserupian effects aligns with recent tests of semi-endogenous growth models that find no effects of population growth on post-1870 productivity in industrialized countries, on England's post-1620 productivity growth, and on recent Asian growth miracles; see Madsen (2008), Madsen et al. (2010), and Ang and Madsen (2011). The previous conclusions also agree with the limited influence of demographic variables on the technological complexity of hunter-gatherer societies and the transition from hunter-gathering to agriculture; see Collard et al. (2013), Fagan (2005), Harlan (1992), Henrich (2004), Kline and Boyd (2010), Read (2006, 2012), and Smith (1995).

The paper unfolds as follows. Section 2 provides some background for the empirical tests of the paper; Section 3 presents tests that use modern population data; Section 4 discusses tests based on pre-modern data and differences in initial population sizes; Section 5 proposes a simple analytical framework to account for the empirical patterns uncovered in the paper; Section 6 briefly concludes.

<sup>&</sup>lt;sup>2</sup>This paper, however, is not a test for Malthusian dynamics. In contrast to Ashraf and Galor (2011), for example, the analysis does not focus exclusively on the pre-modern period. This paper, in fact, provides a clear illustration of the breakdown of Malthusian theory. This paper lacks direct data on technological differences across space and time. Comin et al. (2010) assembled a dataset on the adoption of technology during pre-modern times. The concluding section establishes consistency with their findings.

<sup>&</sup>lt;sup>3</sup>The findings are also relevant due to the renewed interest in global environmental change and population externalities; see, e.g., Acemoglu et al. (2012), Baland and Robinson (2002), Bohn and Stuart (2015), Cohen (1995), Dasgupta (2000), and Lee and Miller (1990).

<sup>&</sup>lt;sup>4</sup>Ravallion (2010) previously discussed the fragility of Kremer's (1993) findings, but in a different context. Ravallion (2010) noted that Kremer's (1993) model has a *spacing implication*: longer time periods between observations imply higher growth rates. Ravallion (2010) showed that the global data contradicts the spacing implication. The analysis presented here complements Ravallion (2010). For example, I examine different predictions and consider additional sources of data; specifically, data with evenly spaced observations.

#### 2 Some theoretical background

This section extends some earlier models of population growth and technological change; particularly, Kremer (1993) and Klasen and Nestmann (2006). I explicitly allow for exogenous determinants of demographic and technological change in order to introduce a competing mechanism into these models; I also derive several new testable predictions.

Let N(t) and A(t) represent population size and the level of technology at date  $t \ge 0$ , with  $N(0) = N_0 > 0$  and  $A(0) = A_0 > 0$ . The aggregate production function is  $Y(t) = A(t)N(t)^{\eta}T^{1-\eta}$  with  $0 < \eta < 1$ , where T is land, in fixed supply. Technological change satisfies

$$\dot{A}(t) = \lambda A(t)^{\phi} N(t)^{\gamma}, \tag{1}$$

where  $\phi$ ,  $\gamma$ , and  $\lambda$  are positive and fixed parameters:  $\phi$  measures the returns to scale to knowledge,  $\gamma$  captures the influence of population on the production of knowledge, and  $\lambda$  represents exogenous influences on technological change;  $\phi$  and  $\gamma$  are associated with the "standing on shoulders" and "fishing out" knowledge externalities; see Jones (2005). (Later, in Section 5, I revisit the formulation of "fishing out" externalities to consider a more general setting.) If  $\phi = 1$  and  $\gamma = 0$ , technology would grow exogenously at a rate  $\lambda$ .

Output per capita is  $y(t) = A(t)(N(t)/T)^{\eta-1}$ . There is an invariant "subsistence level"  $\bar{y}$  with  $y(t) = \bar{y}$  so that

$$\frac{\dot{N}(t)}{N(t)} = \frac{1}{1-\eta} \frac{\dot{A}(t)}{A(t)},$$
(2)

which, once combined with Eq. 1, yields

$$\dot{N}(t) = \theta N(t)^{\alpha},\tag{3}$$

with  $\theta \equiv \lambda \bar{y}^{\phi-1} T^{(1-\eta)(1-\phi)}/(1-\eta)$  and  $\alpha \equiv (1-\eta)(\phi-1)+\gamma$ .

The composite parameter  $\alpha$  is central to the theory. This parameter captures the race between Malthusian and Boserupian effects. If  $\alpha < 1$  because  $\gamma < 1 + (1 - \eta)(1-\phi)$ , Malthusian effects dominate: population growth and size will be negatively related and the economy will converge to a steady-state with constant population size. If  $\alpha > 1$  because  $\gamma > 1 + (1 - \eta)(1 - \phi)$ , Boserupian effects dominate: population growth and size will be positively related and the economy will converge to a steady-state with constant population size. If  $\alpha > 1$  because  $\gamma > 1 + (1 - \eta)(1 - \phi)$ , Boserupian effects dominate: population growth and size will be positively related and the economy will exhibit increasing population growth. Note also that since N(t) is endogenous, to test the theory, one needs to solve Eq. 3 to find the "exogenous" determinants of population. Expression (3) is a Bernoulli differential equation. Let  $N^*(t|\theta, N_0)$  denote its solution as a function of exogenous factors. Then,

$$\ln \left[ N^*(t|\theta, N_0) \right] = \frac{\ln \left[ \theta (1-\alpha)t + N_0^{1-\alpha} \right]}{1-\alpha},$$
(4)

defined for  $t < t^* \equiv N_0^{1-\alpha}/\theta(\alpha - 1) > 0$  when  $\alpha > 1$ . Let  $n^*(t|\theta, N_0)$  denote the growth rate of population along the solution path,

$$n^*(t|\theta, N_0) \equiv \frac{d\ln\left[N^*(t|\theta, N_0)\right]}{dt} = \frac{\theta}{\left[\theta(1-\alpha)t + N_0^{1-\alpha}\right]}.$$
(5)

Expressions (3) and (5) yield the following testable predictions:

(i) If  $\alpha > 1$ , population growth is an increasing function of population size. Moreover, along the solution path, population growth  $n^*(t|\theta, N_0)$  increases with: (ii) exogenous factors,  $\theta$ ; (iii) time, t; and (iv) initial population size,  $N_0$ .

Predictions (i) and (iv) were previously examined by Kremer (1993). Prediction (ii) has not been previously formulated or tested in the literature. Through  $\theta$ , for example, Eqs. 3 and 5 predict a positive effect of exogenous technological factors  $\lambda$  and arable land *T* on the growth rate of population. That is, exogenous factors are predicted to have growth effects and not just level effects on population. Malthusian theory, in contrast, associates exogenous variables to population levels but not to growth rates; see, e.g., Ashraf and Galor (2011). In the limit as Malthusian and Boserupian effects perfectly cancel each other, i.e.,  $\alpha \rightarrow 1$ , the model yields exogenous exponential growth  $n^*(t|\theta, N_0) = \theta$ , which is independent of time and the initial population  $N_0$ . In this special case discussed below, all population growth differences will be the result of differences in exogenous factors.

Prediction (iii) has been previously examined but in a very different context. Ravallion (2010) examined spacing, which is an aspect somewhat related to prediction (iii). Ravallion (2010) interpreted the time index t as the length between observations. Accordingly, a longer time period between observations implies higher population growth rates. Differences in the spacing between observations are specially relevant for very early data in which uneven spacing is prevalent; see, e.g., Kremer (1993, Table 1). I do not focus on spacing because the data sources used here are roughly evenly spaced. Prediction (iii) might seem mechanical. The growth rate of population, however, is predicted to change significantly over time. In particular, population size and the growth rate of population should approach infinity in finite time, i.e.,  $\lim_{t\uparrow t^*} N^*(t|\theta, N_0) = \lim_{t\uparrow t^*} n^*(t|\theta, N_0) = \infty$  in Eqs. 4 and 5, so one should see a rapid acceleration in growth rates as time advances.

The previous predictions also apply to the more general framework of Klasen and Nestmann (2006). Klasen and Nestmann (2006) considered population density as an argument in the production of knowledge, e.g., Eq. 1 becomes  $\dot{A}(t) = \lambda A(t)^{\phi} N(t)^{\gamma} (N(t)/T)^{\sigma}$ , with  $\sigma > 0$ . In this case,  $\theta \equiv \lambda \bar{y}^{\phi-1} T^{(1-\eta)(1-\phi)-\sigma}/(1-\eta)$  and  $\alpha \equiv (\phi-1)(1-\eta) + \gamma + \sigma$ . Including population density in Eq. 1 enhances the relationship between population size and growth in  $\alpha$ , but weakens the relationship between arable land and population growth.

The testable predictions are reduced-form so it is not generally possible to discriminate between the different versions of the knowledge production function. Ideally, one would like to examine the structural equations to identify  $\eta$ ,  $\phi$ ,  $\sigma$ , and  $\gamma$  separately instead of the composite term  $\alpha$ ; this is not feasible due to the lack of data for A(t). For instance, it is not possible to test for specific values of the parameters  $\gamma$  and  $\phi$ , which are important for setting competing endogenous growth theories apart.<sup>5</sup> The value of  $\sigma$  is also useful to discriminate between the previous versions of  $\dot{A}(t)$ . Although it is not possible to separately identify the different factors determining  $\alpha$  and  $\theta$ , I will empirically study the effect of arable land on population growth.

The previous predictions rely on the (Malthusian) assumption that technological change augments population size and leaves income per capita at some subsistence level. Such assumption is less problematic for the pre-modern era since there are several indications of support for the Malthusian model; see, e.g., Ashraf and Galor (2011). Another perhaps more serious concern is the following. In Eqs. 3 and 5, population grows due to exogenous factors, captured by  $\theta$ , and endogenous ones, captured by  $\alpha$ . Thus, to properly identify the relative strength of Boserupian or Malthusian effects based on differences in initial population sizes  $N_0$ , one must control for potentially confounding influences in the form of differences in  $\theta$ . Section 4 discusses this problem in the context of prediction (iv).

#### **3** Tests based on modern population data

This section examines several versions of predictions (i) to (iii) using some well-established data that covers the pre-modern and modern eras. The next section examines prediction (iv) using pre-modern data. I separate these tests because (iv) relies on the isolation of different world regions before the European expansion.

I generally focus on modern population data since existing population estimates for the distant past are severely limited. (I discuss some data concerns in an Appendix *not* for publication.) Several authors have provided longitudinal estimates of national populations in post-agricultural times. I rely on McEvedy and Jones (1985) and Biraben (1979) to study changes within world regions or countries. I use Whitmore et al. (1990) to examine particular agricultural centers. None of the available sources are census measures, and it is recognized that population data are highly uncertain; see, e.g., Caldwell and Schindlmayr (2002) for a critical assessment of past population estimates.<sup>6</sup> McEvedy and Jones (1985) and Biraben (1979), however, are independent sources that cover a large number of geographic areas: McEvedy and Jones (1985) contains 73 separate countries, and data from 200

<sup>&</sup>lt;sup>5</sup>For example, it is not possible to separately test for "scale effects" (e.g.,  $\gamma = 1$  and  $\phi = 0$ ) and "market size effects" (e.g.,  $\gamma < 1$  and  $\phi < 1$ ). Madsen (2008) used patents and R&D data for OECD economies to examine Schumpeterian and semi-endogenous versions of the knowledge production function. These tests require data that is not available for pre-modern times. Madsen et al. (2010) implemented tests for the shape of the knowledge production function during the Industrial Revolution, with data that cannot be extended far back in time. See also Ang and Madsen (2011) for similar tests in high performing Asian economies.

<sup>&</sup>lt;sup>6</sup>There are multiple alternative estimates of past population sizes including Clark (1967), Bairoch (1988), and more recently Maddison (2001). Caldwell and Schindlmayr (2002) discusses in detail the connections between these alternative estimates and the sources used here.

BC to 1975. Biraben (1979) contains 12 regions, and population data from 400 BC to 1970. In both sources, spacing is more or less even although the frequency of observations increases slightly at the end of the sample. For convenience, I will refer to the world regions in Biraben (1979) simply as countries. Whitmore et al. (1990) provide archeological reconstructions for agricultural centers. I use data from local agricultural centers primarily for sensitivity purposes.

#### 3.1 Tests of prediction (i)

**Population growth and population size.** Let  $n_{i,t}$  be the geometric average growth rate of the population for country *i*, between periods t - 1 and *t*. (The geometric average is more appropriate than the arithmetic average for describing population growth.) I consider the following simple econometric model:

$$n_{i,t} = \beta_0 + \beta N_{i,t-1} + \delta_i + \varepsilon_{i,t},\tag{6}$$

where  $N_{i,t-1}$  denotes the population size of country *i* in period t - 1, and  $\delta_i$  denotes an unobserved country-specific component that influences population growth. The error term,  $\varepsilon_{i,t}$ , captures all other omitted factors, and  $\beta_0$  is a constant term. I consider both random and fixed effects but I only report the fixed effects (FE) estimates since the findings are invariant to the choice of specification.

Prediction (i) requires  $\beta > 0$ . I first display the resulting tests graphically. Figure 1 plots population size and the population growth rate from 1 MYA through 2010 based



Fig. 1 World population size and its growth rate, 1 MYA to 2010



Fig. 2 World population size and its growth rate, 1 MYA to 1650

on global data from Kremer (1993) and Ravallion (2010).<sup>7</sup> Two points are important about this figure. First, population growth and size are positively related in the complete sample. Second, this positive relationship seems heavily influenced by the modern demographic transition: data prior to 1600 are dwarfed in Fig. 1 so one needs a smaller scale to make any pre-1600 pattern visible, as in Fig. 2 below; and, post-1960 data, updated by Ravallion (2010), show a negative association between population growth and size.

In terms of point estimates for  $\beta$ , Table 1, panel A, estimates Eq. 6 for the entire sample and the three global population data, using OLS. Column (1) reproduces the estimates in Kremer (1993). Column (2) uses data from McEvedy and Jones (1985), and Column (4), data from Biraben (1979). For all data sources, the point estimates for  $\beta$  are positive and significant. Indeed, since the data used to produce the point estimates in (2) and (4) covers a more recent sample, one should expect their point estimates to exceed those of column (1). As expected, the point estimates in columns (2) and (4) are marginally greater than those that rely on data that begins 1 MYA (column (1)).

The previous estimates are based on global samples. Column (3) in Table 1, Panel A, presents estimates of Eq. 6 using country data from McEvedy and Jones (1985). Column (5) presents the same estimates using country data from Biraben (1979). I

<sup>&</sup>lt;sup>7</sup>There are very few alternatives sources to estimate the human population in the distant past; see, e.g., Hassan (1981). Deevey (1960) is a common source to most estimates. Deevey's (1960, p. 195) estimates are available "from the inception of the hominid line one million years ago," but his population data is so speculative that Deevey (1968, p. 248) himself remarked: "my own treatment of this, published some years ago in *Scientific American*, was not very professional." Deevey's (1960) data is especially problematic because it is biased toward accepting hypotheses (i) to (iii). I discuss this point in an Appendix not for publication.

only report the fixed effects estimates; random effects estimates are available upon request. The point estimates of  $\beta$  in these specifications are positive, statistically significant, and virtually identical across data sets and specifications. Thus, while the theoretical framework applies on a global scale, the empirical findings hold on a much more disaggregated setting. In fact, the point estimates for  $\beta$  in columns (3) and (5) are orders of magnitude greater than those for the world population. The fixed-effects estimates, for instance, are about five times larger than those in (2) and (4). To the extent that fixed-effects controls for omitted variables, one should prefer the country-level estimates over the global OLS estimates. (This difference between the global and country-level sample is somewhat surprising because fixed-effects estimators typically produce lower estimates than OLS. The country-level estimates are larger here because the timing of the modern demographic transition differed across countries. Differences in timing imply a lower covariance between population growth and population size in global samples, but a higher covariance between these variables at the country level.)

Overall, Table 1, Panel A, shows that population growth rates are strongly correlated with population size, as predicted by (i). If anything, the correlations in country-level data and alternative specifications are actually stronger that those previously documented in the literature.

**Log-population growth and log-population size** The OLS estimates of Eq. 6 speak about the value of  $\alpha$  in Eq. 3 only as an approximation, e.g., one can approximate (3) by  $n(t) \approx \theta(1 - \alpha) + \alpha \theta N(t)$ . The relevant reduced-form elasticity  $\alpha$  can be more directly estimated using a log-log specification from Eq. 3:

$$\ln n_{i,t} = \rho_0 + \rho \ln N_{i,t-1} + \delta_i + \varepsilon_{i,t}.$$
(7)

This log-log specification, however, is inconsistent with zero or negative population growth. In the world population estimates in Kremer (1993, Table 1), there is only one instance of a population decline, namely the Black Death, but several instances of zero growth. To deal with nonpositive growth rates, I first estimate Eq. 7 using positive growth rates only and treat  $1 + \rho$  as an *upper bound* for  $\alpha$ .<sup>8</sup> Then, as part of the sensitivity analysis, I will assess the bias in  $\rho$  relative to  $\alpha$  using long-term demographic data based on archeological reconstructions for well-defined agricultural centers with documented negative growth rates; see Whitmore et al. (1990).

Table 1, Panel B, presents the estimates for Eq. 7 under the same specifications of Panel A, once all nonpositive growth rates are dropped from the samples. All point estimates are positive and significant. As a useful benchmark, a point estimate of  $\hat{\rho} = 1$  implies that population growth is directly proportional to population size, i.e.,  $\hat{\alpha} = 2$  in (3). Global data suggest that  $\hat{\rho} = 1$  might not be a bad approximation. For example, with Kremer's (1993) data,  $\hat{\rho} = 0.80$  which implies that  $\hat{\alpha} = 1.8$ . The

<sup>&</sup>lt;sup>8</sup>I re-scaled the growth rates so that all values of the normalized population growth rates are positive. The results are sensitive to the normalization I used. I do not present these results here, but as expected,  $\rho$  is an upper bound. I also considered a nonlinear OLS estimation of the model in the form of (4) but the estimates did not converge or were too sensitive to the initial guess to be of any value.

	I. Kremer (1993) 1 MYA to 1990	II. McEvedy 200 BC to 19	and Jones (1985) 75	III. Biraben (1979) 400 BC to 1970		
	World population	World population	Country data	World population	Country data	
	(1)	(2)	FE (3)	(4)	FE (5)	
A. Depende	nt variable is population	n growth				
$N_{i,t-1}$	5.08***	7.06***	36.3***	6.99***	39.3 ***	
.,.	(0.55)	(0.57)	(7.99)	(0.68)	(6.98)	
Obs.	37	20	679	25	300	
Countries	_	_	73	_	12	
$R^2$	0.90	0.93	0.05	0.87	0.11	
B. Depende	nt variable is log-popul	ation growth				
$\ln N_{i,t-1}$	0.80***	1.30***	0.46***	1.48***	0.57 ***	
	(0.06)	(0.10)	(0.04)	(0.31)	(0.06)	
Obs.	33	16	566	21	176	
Countries	-	_	73	-	12	
$R^2$	0.86	0.90	0.16	0.52	0.45	

Table 1 Population growth and popula	ation	size
--------------------------------------	-------	------

Robust standard errors are in parentheses.  $N_{i,t-1}$  denotes population for country *i* in period t - 1. Panel B dropped the non-positive growth rates. Point estimates and standard errors in Panel A are multiplied by  $10^5$  to aid visually with the presentation of the results. \*\*\* and \*\*denote significance at the 1 and 5 percent levels.

global estimates obtained with McEvedy and Jones (1985) and Biraben (1979) are larger than in Kremer (1993) because these samples start at later dates. Table 1, Panel B, also shows that the country-level estimates of  $\rho$  are generally lower than the global estimates. Using country data, the relevant elasticity is  $\hat{\alpha} = 1.5$ , a value still larger than one. (Although several observations are lost in country-level samples, the log transformation stabilizes the variance of demographic data and that is a likely reason for the differences between country and global estimates in Table 1, Panel B.)

**Sensitivity analyses** Table 1 shows that the point estimates for  $\beta$  and  $\rho$  increase as the initial sample date increases. I next examine changes in these estimates when recent observations are excluded from the sample. The logic behind these exercises is simply that recent observations are likely to be very influential. A variety of diagnostics can be used to detect influential observations; see Chatterjee and Hadi (1988). *High-leverage* diagnoses atypical observations. Leverage also determines the *Cook's distance* diagnostic, which measures the influence of a given observation on the point estimates. For Table 1, Panel A, the post-1900 observations in McEvedy and Jones (1985) and Biraben (1979) and the post-1940 observations in Kremer (1993) are

high-leverage points. That is, the population size values during these periods are atypical compared to the majority of the sample. The Cook's distances also suggest that the point estimates in Eq. 6 are sensitive to the removal of observations from these periods.<sup>9</sup>

How much would the point estimates for  $\beta$  change if more recent observations are excluded from the samples in Table 1? Recall from Table 1, Panel A, that the point estimate from Kremer (1993) is 5.08 (s.e. 0.5). This point estimate declines to 4.22 and 4.56 if post-1900 and post-1800 observations are excluded, respectively. These estimates are still significant. If post-1700 observations are excluded, however,  $\beta$  declines to 2.60 (s.e. 1.14) and becomes only marginally significant; in fact, if post-1600 observations are excluded,  $\beta$  declines to 1.75 (s.e. 1.14) and is no longer significant. As Table 2 shows the estimates for  $\beta$  in the alternative global samples of McEvedy and Jones (1985) and Biraben (1979) are also insignificant if post-1800 observations are excluded.

The lack of significance in the relationship between population growth and size in global samples is due to a large decline in the point estimate for  $\beta$ , not only to an increase in the standard errors. Figure 2 shows, for example, that there is no systematic relationship between population growth and size in pre-modern samples.

One can question the relevance of a sensitivity analysis for global data because the degrees of freedom in the global samples are considerably reduced; also, the small global sample makes influential observations more likely to occur naturally. A decline in the point estimates of  $\beta$ , however, is evident even in the country data analyzed in Table 2. Excluding post-1900 observations has a dramatic impact on the point estimates for country data. All point estimates, with the exception of the fixed effects in Biraben (1979), which is only marginally significant, become insignificant. Excluding post-1800 observations has a similar effect. Only the fixed effects estimate in McEvedy and Jones (1985) is significantly different from zero. Finally, no point estimate is significant if post-1700 observations are excluded. In these samples, the estimates of  $\beta$  have the "wrong" (negative) sign. That is, these estimates suggest that population growth and size are *negatively* rather than positively associated. If one excludes post-1600 observations, some of these negative estimates actually become significant: i.e.,  $\hat{\beta} = -29.80$  (s.e. 15) in Biraben (1979).

As in the global sample, the decline in the significance of the point estimates for the country data is not primarily due to an increase in the standard errors, but to a decline in the point estimates themselves. This fact is important because the most recent data is likely to be of the highest quality hence one would expect precision to be lost if these observations are removed. As Table 2 shows, this is not the case: the

<sup>&</sup>lt;sup>9</sup>An observation is considered high leverage if its leverage exceeds 4 /N.obs; see, Chatterjee and Hadi (1988, p. 100). All of Cook's distances (D) should be roughly equal. A relatively large Cook's distance indicates an influential observation. I use the cut-off values based on D > F(0.5, 2, N.obs - 2); see, Chatterjee and Hadi (1988, p. 119). Influential observations are not necessarily outliers, but their inclusion is likely to influence the estimation of the regression coefficients. For example, because OLS minimizes square deviations, the estimates place a relatively heavy weight on atypical observations. I also computed the previous diagnostics for country data. Leverage and the Cook's distance show a significant positive time trend. This indicates that recent observations are more influential than pre-modern observations.

	World	Country	World	Country	World	Country	World	Country
	pop.	data	pop.	data	pop.	data	pop.	data
	OLS	FE	OLS	FE	OLS	FE	OLS	FE
	(1)	(2)	(1)	(2)	(1)	(2)	(1)	(2)
I. McEvedy	and Jones (	1985)						
	200 BC to	1900	200 BC to	1800	200 BC	to 1700	200 BC	to 1600
$N_{i,t-1}$	6.06***	12.90	5.38***	20.10**	2.24	-2.90	3.93	-7.18
	(0.61)	(9.24)	(1.59)	(9.75)	(2.13)	(10.50)	(3.01)	(16.00)
Obs.	18	533	16	404	14	346	12	302
Countries	-	68	-	63	-	63	-	63
II. Biraben (	1979)							
	400 BC to	1900	400 BC to	1800	400 BC	to 1700	400 BC	to 1600
$N_{i,t-1}$	5.13***	13.50*	4.06**	4.29	0.82	-19.40	-0.34	-29.80*
	(0.64)	(7.42)	(1.66)	(10.30)	(2.68)	(11.80)	(3.91)	(15.40)
Obs.	23	276	21	252	19	228	18	216
Countries	-	12	-	12	-	12	-	12

 Table 2 Population growth and population size. Alternative sample periods

The dependent variable is population growth.  $N_{i,t-1}$  denotes population for country *i* in period t-1. The specifications are the same as in Table 1. Robust standard errors in parentheses. Point estimates and standard errors multiplied by 10<sup>5</sup> to aid visually with the presentation of the results. \*\*\*, \*\*, and \*denote significance at the 1, 5, and 10 percent level.

point estimates are not becoming more "imprecise". A positive relationship between population growth and size, instead, is mostly a feature of the modern era.<sup>10</sup> Finally, I have considered changes to  $\beta$ , but the point estimates for  $\rho$  in the log-log specification (7) change qualitatively in similar ways as in Table 2. Estimates available upon request show that  $\hat{\rho}$  declines considerably, although the significance is not always lost. The fact that some of the point estimates of  $\rho$  remain significant has to be taken with a grain of salt because nonpositive growth rates have been excluded. In general, if post-1600 data is excluded, the upper bound estimate  $\rho$  is reduced in half or more.

**Local demographic data** This subsection briefly presents an additional perspective of the relationship between population growth and size based on data from demographic reconstructions in local areas where archaeological material covers a long span of time; see Whitmore et al. (1990). The local areas are the Tigris-Euphrates lowlands, the Egyptian Nile Valley, the basin of Mexico, and the central Maya lowlands. Two of these areas are located in the Old World and two in the New World

<sup>&</sup>lt;sup>10</sup>Focusing only on modern observations, an alternative way to examine the sensitivity of the results, agrees with the findings of Table 2. To save space, I only discuss the post-1600 estimates of Eq. 6 for the global data and the fixed effects regional estimates in McEvedy and Jones (1985), where more data is available. For the global data,  $\beta = 7.52$  (s.e. 0.70). For the fixed effects estimates,  $\beta = 42.60$  (s.e. 11.70). These estimates are positive, significant, and larger than the estimates for the entire sample. These results and Table 2 show that the strong positive relationship in the recent samples drives the results in Table 1.

thus producing a general geographic coverage. The population data in these areas begins at about 6 KYA (for some local areas), and this considerably extends the sample period. A local perspective does not intend to substitute the previous discussion based on global and country data, but to complement it with less speculative population measurements for more homogeneous areas and longer periods of time. Since the theory is designed to examine global data, its predictions for local populations are not fully specified. For instance, aggregation might make relevant local patterns irrelevant at the aggregate level, especially if the timing of population expansions and declines in local areas tends to covary negatively. To counter this concern, notice that the reconstructions are done for some of the most technologically advanced societies of the pre-modern era.

Figure 3 reproduces Figs. 1 and 2 using data for the previous four local areas. I only examine the local data graphically, but include regression estimates in the relevant figures. Each panel of the figure presents one local area and three linear trends. The baseline trend uses all data points and the entire sample period. The second trend (positive growth only) omits nonpositive growth rates, as a way to assess the bias in  $\rho$  relative to  $\alpha$ , and the third trend uses only pre-1900 data, as a way to assess the sensitivity of the estimate to changes in the end of the sample.

The patterns of population in Fig. 3 are considerably more diverse than in the global and country data. With the exception of Panel (d), all figures show a baseline positive association between population size and growth; Panel (b) actually shows a stronger association between population size and growth in the pre-1900 sample. (In the Whitmore et al. (1990) data, the Nile valley experienced a sharp population decline between 1420 AD and 1600 AD and a rapid surge afterwards. This "cycle" is likely the reason for the steeper pre-1900 trend in Fig 3b.) The Mayan low-lands, Panel (d), have experienced a steady population decline since 800 AD to the point that this region has not yet reached a population size near its peak during the pre-modern era. Panels (a) and (c) roughly agree with the global and country-level patterns in the sense that recent observations are the main reason for the baseline positive association between population size and growth. In these panels, for example, once post-1900 data is omitted, the association between these variables becomes significantly negative.

Figure 3 also shows numerous episodes of negative growth. The magnitude of the bias in  $\rho$  relative to  $\alpha$  depends on the frequency of negative growth rates in the data. Panels (a) and (b) feature relatively small negative growth rates and therefore they suggest a relatively small bias in the log-log specification. In contrast, panel (c) features large negative growth rates of population and therefore a large bias. The Basin of Mexico in panel (c), in fact, shows that the relationship between population size and its growth rate is very sensitive to the sample examined and the econometric specification.

**Overview** There is a positive association between population growth and size in global and country-level samples. This association, however, is unstable. Point estimates for  $\beta$  lose significance when recent observations are excluded from the sample. In local data for key agricultural centers in the Old and New Worlds, there are more



Fig. 3 Local population size and growth rates

varied patterns of population growth but recent observations also have a disproportionate influence on the relationship between population growth and size. The instability in the relationship between population growth and size is due to the transient acceleration in population growth during the modern demographic transition. This instability is thus evidence against an underlying long-term Boserupian influence. Indeed, the 1960s marked the end of the demographic transition for most developed countries. In post-1960 samples, the relationship between population growth and size is negative (Fig. 1).

#### 3.2 Tests of prediction (ii)

Recall from Eq. 4 that exogenous factors associated with technological change and arable land influence population growth. These growth effects have not been previously examined by the literature, although there is a clear (Malthusian) connection between technological conditions and the level of population across world regions during the pre-modern era; see, e.g., Ashraf and Galor (2011).

To test for growth effects, I consider the following econometric model:

$$n_{i,t} = \varphi_0 + \varphi_\theta \theta_i + \varphi_x X_{i,t} + \varepsilon_{i,t}, \tag{8}$$

where  $\theta_i$  denotes time-invariant exogenous factors, and  $X_{i,t}$  denotes the control variables. More specifically, I use time, t, and population size,  $N_{i,t-1}$ , as control variables. Since  $\theta_i$  is time-invariant, I cannot rely on fixed-effects. Instead, I present pooled OLS estimates. (The random effects estimates are virtually identical, so to save space, I do not report them.) For these estimates, population size controls for regional-specific and time-varying factors. Furthermore, since I assume that global arable land and other exogenous factors remain constant, I am only able to examine regional data. Finally, I examine McEvedy and Jones (1985), as their data contains a larger number of regions and thus permits a better matching with additional sources.

In a long-term perspective, it is difficult to find "exogenous" variables; it is also difficult to justify time-invariance in  $\theta_i$ . I follow two different approaches. First, I examine the relationship between population growth and arable land. Arable land, as reported in McEvedy and Jones (1985), represents potentially cultivable land. Land is measured in modern times and excludes primarily desert, inland water, and tundra. (I also use total land but the results are unchanged so I do not report these estimates.) Second, I examine biogeographical variables, such as those emphasized by Diamond (1997) and coded (at the regional/country level) by Hibbs and Olsson (2004). Country boundaries are endogenous so arable land is not necessarily exogenous. Concerns with the endogeneity of country boundaries, however, can be minimized because there is a high degree of spatial correlation in economic and demographic patterns. Biogeographic variables are often taken as exogenous, although they likely experienced temporal changes due to extinctions, for example. Since there are no longitudinal estimates of biogeographic conditions, I assume that  $\theta_i$  is time invariant.

**Population growth and arable land** To provide a more specific context for the point estimates of  $\varphi_{\theta}$  in Eq. 8, consider the special case of  $\alpha = 1$ , when population grows exogenously at a rate  $n^*(t|\theta, N_0) = \lambda \bar{y}^{\phi-1}T^{(1-\eta)(1-\phi)}/(1-\eta)$ ; see Eq. 5. The growth rate of population can be approximated by  $n^*(t|\theta, N_0) \simeq (1-\eta)(1-\phi) \ln T + \ln[\lambda \bar{y}^{\phi-1}/(1-\eta)]$ . The point estimate for the relationship between population growth and log-land is  $\hat{\varphi}_{\theta} \equiv (1-\eta)(1-\phi)$ . If  $\alpha = 1$ , then some simple substitutions imply that  $\hat{\varphi}_{\theta} = \gamma - 1$ , whose sign is informative about the role of population size on the knowledge production function (1). In particular,  $\hat{\varphi}_{\theta} > 0$  is consistent with Boserupian effects.

Table 3, Panel A, includes log-arable land as a predictor of the growth rate of population. (Total land yields similar results and for convenience I do not present those results. Since population and time are strongly correlated, I do not include both factors simultaneously as controls.) Column (1) includes no other controls, column (2) includes a time trend, and column (3) controls for population size. I consider the complete sample first. The point estimate in column (1) is  $\hat{\varphi}_{\theta} = 9.7$ , which is statistically significant. As I just noted, this point estimate suggests positive (and large) effects of population size on the production of knowledge. Moreover, the inclusion of time or population size only changes the value of  $\varphi_{\theta}$  marginally, without lowering its statistical significance. In column (3), the point estimate is  $\hat{\varphi}_{\theta} = 10.66$ , which is not different from the one in column (1).

	200 BC to 1970			200 BC to 1600			
	(1)	(2)	(3)	(1)	(2)	(3)	
A. Pooled OLS	5: Arable land (i	n km <sup>2</sup> )					
Log arable	9.70***	6.31***	10.66***	-1.53**	-1.68**	-1.90**	
land	(2.21)	(1.95)	(2.45)	(0.69)	(0.69)	(0.83)	
Time trend	No	Yes	No	No	Yes	No	
Population	No	No	Yes	No	No	Yes	
Obs.	679	679	679	302	302	679	
B. Pooled OLS	S: Biogeography						
Number of	-0.33	0.17	0.60	-3.32**	-3.85***	-0.61***	
plants	(0.36)	(0.33)	(0.42)	(1.44)	(1.49)	(0.17)	
Number of	-12.88***	- 9.19***	-13.81***	2.18**	2.99 ***	3.16***	
animals	(1.91)	(1.86)	(2.04)	(0.88)	(0.94)	(0.96)	
Time trend	No	Yes	No	No	Yes	No	
Population	No	No	Yes	No	No	Yes	
Obs.	573	573	573	270	270	270	

Table 3 Population growth, land, and biogeography. Alternative sample periods

The dependent variable is population growth. All specifications are pooled OLS. Robust standard errors are in parentheses. Point estimates and standard errors for arable land are multiplied by  $10^3$  to aid visually with the presentation of the results. \*\*\*, \*\* , and \*denote significance at the 1, 5, and 10 percent levels.

Panel A shows that in the complete sample, population grew at faster rates in larger geographical areas, as predicted by (ii). The findings in the alternative sub-samples, however, provide a very different view. In the sample ending in 1800 (which I do not report for convenience), arable land is no longer significant, with or without a time trend or population controls. In the sample ending in 1600, which is reported in Table 3, the estimate of  $\varphi_{\theta}$  has the "wrong" (negative) sign, and it is stable and significant. The point estimates in these specifications are statistically the same. The reversal in the point estimate of  $\varphi_{\theta}$  implies that up until 1600, population growth was slower in larger geographic areas. In the special case of  $\alpha = 1$ , a negative point estimate implies that population size had a *negative* effect on the production of knowledge during the pre-modern era.

**Population growth and biogeography** Table 3, Panel B, presents a complementary strategy. I matched the countries in McEvedy and Jones (1985) with the biogeographic information coded by Hibbs and Olsson (2004). Table 3, Panel B, examines the influence of the *number of plants*, which is the average numbers of locally available wild plants suited for domestication 12 KYA in various parts of the world, and the *number of animals*, which is the number of species of wild terrestrial herbivore

	World population	Country data	World population	Country data	World population	Country data
	OLS	FE	OLS	FE	OLS	FE
	(1)	(2)	(1)	(2)	(1)	(2)
I. McEvedy	and Jones (1985)					
	200 BC to 19	70	200 BC to 18	00	200 BC to 1600	
Time	4.09**	5.01***	1.34**	1.42***	0.74	0.80 ***
	(1.51)	(0.49)	(0.51)	(0.22)	(0.47)	(0.24)
Obs.	20	679	18	404	12	302
Countries	-	73	-	63	-	63
II. Biraben (	1979)					
	400 BC to 19	75	400 BC to 1800		400 BC to 1600	
Time	3.52***	4.07***	1.00	0.60**	0.37	0.11
	(1.43)	(0.58)	(6.14)	(0.30)	(0.69)	(0.31)
Obs.	25	300	21	252	18	216
Countries	-	12	-	12	-	12

 Table 4
 Trends in population growth. Alternative sample periods

The dependent variable is population growth. t denotes calendar time. The specifications are the same as in Table 1. Robust standard errors are in parentheses. Point estimates and standard errors are multiplied by  $10^5$  to aid visually with the presentation of the results. \*\*\*, \*\*, and \*denote significance at the 1, 5, and 10 percent levels.

and omnivore mammals suitable for domestication in various parts of the world. I consider the same specifications as in Panel A.<sup>11</sup>

Column (1) shows that the number of plants has no predictive power for population growth in the complete sample. The number of animals is negatively and significantly associated with population growth; a large number of domesticable animals lowers population growth in the complete sample. The important observation is that the variation of the point estimates across sub-samples is the same as in Panel A. In the sample ending in 1800 (not displayed), none of the biogeographic variables is significant whereas in the sample ending in 1600, displayed in Table 3, Panel B, the signs of the point estimates  $\hat{\varphi}_{\theta}$  are completely reversed. In the pre-1600 sample, domesticable animals have a positive and significant association with population growth and the number of plants has a negative and significant association with population growth. As in Panel A, adding a time trend or population size as a control does not alter the previous conclusions.

<sup>&</sup>lt;sup>11</sup>The number of domesticable plants and the number of animals are highly correlated in the sample (the correlation coefficient is 0.87). I included the continental axis of orientation from Hibbs and Olsson (2004) as an alternative specification. The results are similar to Panel B and are available upon request.

**Overview** The relationship between population growth and exogenous factors such as arable land and biogeography is also unstable. Predictions relating population growth to arable land support Boserupian views but only in the samples that include modern observations; in pre-modern observations, the relationship between these variables displays a "wrong" (e.g., negative) sign. Biogeography is unimportant in the modern data, but there is a strong association between biogeographic variables and population growth during the pre-modern era. Importantly, as in the case of arable land, the estimated coefficients for the relationship between population growth and biogeography switch signs in the pre-modern samples.

Differential timing for the demographic transition rationalizes the reversals in the relationship between population growth and exogenous factors. The more recent and rapid demographic transition of large developing countries such as India and China would induce a positive effect of arable land on population growth in the complete sample. In contrast, European and Neo-European regions experienced an early onset of the demographic transition. These regions were not large or populous at that time. Differences in timing and initial population size thus imply that one should see a negative association between land and population growth in the premodern sample and a positive association when recent observations are included. The reversal in the role of biogeography can be similarly explained. The rapid early population growth in European and Neo-European regions cannot be a consequence of favorable biogeographic conditions. These regions (i.e., North America and Australia) lacked the conditions for settled agriculture. Yet, their populations transitioned before those of highly-advanced agricultural areas such as China and the near East.

Reversals on the role of biogeographic endowments and arable land on the growth rate of population are consistent with the *reversal of fortune* experienced by European colonies, documented by Acemoglu et al. (2002). The findings presented here, however, hold across all countries and apply to population growth rates not to population densities or the rate of urbanization, which are the outcomes studied by Acemoglu et al. (2002).

#### 3.3 Tests of prediction (iii)

Prediction (iii) implies a time trend in the growth rate of population. For completeness, Table 4 presents estimates of a linear time trend in the global and country samples. As before, I consider alternative sample periods and specifications. The first panel includes the entire sample period. In this panel, and for both data sources, the growth rate of population exhibits a positive time trend, as expected from prediction (iii). The point estimates for the linear trend are similar across specifications and sources, and between the global and country-level samples.

The second and third panels consider samples that end in 1800 and 1600, respectively. There are two notable features in Table 4. First, the point estimates for the trend decline uniformly across the samples. Second, the decline is shared by the estimates based on global and country-level data. The significance of the point estimates also declines as the end point of the sample is reduced. In Biraben (1979), no single estimate is significantly different from zero if post-1600 data are excluded. In McEvedy and Jones (1985), the country estimates remain significant, but the findings do not show any trend in world population. Finally, notice that as in Tables 2 and 3, the decline in significance for the time trend is not due to an increase in the estimated standard errors.

## 4 Tests based on initial population sizes

The empirical analysis has so far focused on the time series of population growth. This section provides an assessment of Eq. 4, or prediction (iv), using the melting of the ice caps that divided the continents following the last glaciation.

The nature of the test The ideal experiment to test prediction (iv) would randomly divide human populations into two groups. These groups would only differ in terms of their initial population size and their populations would remain isolated for a very long period of time. In theory, one should then see a lower level of technological sophistication in the group with the smallest initial population size.

The melting of the ice caps shares many features with the ideal experiment. The land bridges that connected the New World with Asia and New Guinea with the Australian mainland (also Tasmania and Flinders Island) disappeared some 12 KYA, leaving certain populations geographically isolated. Geographic isolation thus created relevant differences in population sizes across various regions in the world; see, e.g., Kremer (1993).

There is, however, a critical limitation. For this natural experiment to provide an unbiased test, one needs the two groups to be similar in terms of exogenous characteristics represented by  $\theta$ . Failure to account for differences in exogenous factors will bias tests of prediction (iv). Consider, for example, a comparison between the New and the Old Worlds. Pre-isolation population densities are assumed to be equal so the New World's smaller area translates into a smaller initial population. The New World was less sophisticated and less densely populated around 1500 AD. This comparison thus suggests that a smaller initial population size hindered the New World's long-term development. The critical limitation is that large geographic areas also have larger numbers of animals and plants suitable for domestication; see, e.g., Diamond (1997). Since biogeography determines population growth (both in theory; see, e.g., expression (4), and in practice; see, e.g., Table 3), a comparison that fails to control for these exogenous differences will incorrectly favor prediction (iv).

Many other regions remained isolated from the Old World until the European expansion. Their technological differences, however, seem largely dominated by non-demographic factors. Mainland Australia and Tasmania (even Flinders Island) were also cut off from the rest of the Old World when the Bassian plain flooded. The size of their land areas is considerably different so one should expect differences in their initial population sizes some 12 KYA. One should also expect significant differences

in the level of technology between these regions around 1500 AD. Kremer (1993, p. 709) and Diamond (1997, p. 313) described some technological differences that have been systematically studied by anthropologists.<sup>12</sup> None of these areas, however, relied on agriculture at the time of the European expansion because there were no animals or plants suitable for domestication. Before the European expansion, all inhabitants of these areas lived as hunter-gatherers.

**Some econometric estimates** This sub-section verifies that the previous discussion is relevant for thinking about systematic tests of prediction (iv). I now examine the growth of regional populations in samples ending in 1500 AD using the following specification:

$$\ln[N_{i,t}] = \pi_0 + \pi_1 \ln[N_{i,0}] + \pi_2 \operatorname{Time} + \pi_3 \{\ln[N_{i,0}] \times \operatorname{Time}\} + \pi_x X_{i,t} + \delta_i + \varepsilon_{i,t}, \quad (9)$$

where  $\ln[N_{i,0}]$  is the log of population in area *i* at time t = 0, during the first period in the sample. (As a robustness check, later on, I replace this variable with arable land.) The term  $\ln[N_{i,0}] \times \text{Time}$  is the cross-product of the initial population and time. As before,  $\delta_i$  is a region-specific unobserved component capturing invariant characteristics in the various regions, and  $X_{i,t}$  are additional control variables. In expression (9),  $\pi_1$  captures how initial differences in population size yield differences in final population size;  $\pi_2$  captures the time trend in population, and  $\pi_3$  captures the double difference between time and the initial population size. The double-difference estimates whether or not areas with larger initial populations grew at faster rates than areas with smaller initial populations. To be consistent with prediction (iv),  $\pi_3$  should be positive.

Table 5 presents the results using data from McEvedy and Jones (1985) and Biraben (1979), under two specifications: pooled OLS and fixed effects. I also present two strategies aimed at accounting explicitly for a possible correlation across the error term due to geographic isolation. I consider robust standard errors and clustering at the continental level. Clustering implies that unobserved shocks during the initial period will have a common spillover in each continent during the remaining periods.

Column (1) includes the interaction term  $\ln[N_{i,0}] \times \text{Time}$  with no other covariates. In column (1),  $\pi_3$  is positive, significant, and virtually identical across data sets. These estimates suggest that larger initial population sizes imply a faster population growth, as prediction (iv) suggests. Columns (2) and (3) include a time trend and the initial population size of the country as controls. Time trends have little impact on the point estimates. Initial population size, however, completely overturns the

<sup>&</sup>lt;sup>12</sup>Henrich (2004) proposed an analytical model of knowledge accumulation and diffusion much in line with Boserupian ideas and used demography to account for Tasmania's technological conditions during pre-modern times. Read (2006) provides a critical assessment of Henrich's (2004) theory and finds population to be a second-order influence. The analytical and empirical relationship between population/group size and cultural complexity in hunter-gather societies is the subject of a considerable literature in anthropology; see, e.g., Collard et al. (2013), Kline and Boyd (2010), and Read (2012). Existing empirical findings seem heavily constrained by small samples and by the fact that hunter-gatherers face different environments that require different technological adaptations and risk strategies.

	OLS	FE			
	(1)	(2)	(3)	(4)	(5)
I. McEvedy and Jones	$(1985), t_0 = 200$	BC			
$\ln[N_{i,0}] \times \text{Time}$	5.25***	5.12***	- 2.61***	-2.58***	-2.56***
	(0.51)	(0.46)	(0.64)	(0.46)	(0.46)
	[0.56]	[0.60]	[0.14]	[0.17]	[0.09]
Time	_	4.53***	8.22 ***	8.51***	7.97***
	(1.25)	(0.81)	(0.78)	(0.63)	
$\ln[N_{i,0}]$	_	_	0.98***	0.95***	-
			(0.04)	(0.06)	
Continental control	No	No	No	Yes	-
Obs/Countries	252	252	252	252	252/22
II. Biraben (1979), $t_0 =$	400 BC				
$\ln[N_{i,0}] \times \text{Time}$	4.54***	5.24***	-2.05***	-2.05***	-2.05***
	(0.32)	(0.33)	(0.50)	(0.50)	(0.42)
	[1.05]	[1.07]	[1.17]	[1.18]	[1.16]
Time	_	-4.56***	9.36***	9.36***	9.36***
		(1.41)	(0.11)	(0.10)	(0.99)
$\ln[N_{i,0}]$	_	_	1.04***	0.91***	-
			(0.04)	(0.05)	
Continental control	No	No	No	Yes	-
Obs/Countries	216	216	216	216	216/12

 Table 5
 Population change and initial population before 1500

The dependent variable is log population size. Robust standard errors are in parentheses. In brackets are standard errors clustered at the continental level. The point estimates and s.e. for interaction between initial population and time,  $\ln[N_{i,0}] \times \text{Time}$ , have been multiplied by  $10^4$  to aid visually with the presentation of the results. The point estimates and s.e. for Time have been multiplied by  $10^3$ . \*\*\*, \*\* , and \*denote significance at the 1, 5, and 10 percent levels based on the robust standard errors.

point estimates of  $\pi_3$ . In column (3),  $\pi_3$  is no longer positive; it is negative and statistically significant. Column (4) includes continental controls for Africa, America, Asia, Europe, and Australia. The point estimates for  $\pi_3$  remain virtually unchanged, although (as expected due to the control for geographic region) the clustered standard errors yield a reduced significance compared to the robust standard errors. Column (5) presents the fixed effects estimates (random effects estimates are virtually identical to the fixed effect ones so I omit them). In these specifications,  $\pi_3$  is also negative and statistically significant.<sup>13</sup>

Table 6 reproduces Table 5 for alternative specifications, for the McEvedy and Jones (1985) data. First, I use 1 AD (or year 1) as the starting date rather than 200

<sup>&</sup>lt;sup>13</sup>I also added dummy controls for the Black Death and the Mongol invasions to specification (5) in Table 5. The point estimates for these events are negative, but they do not alter the estimate of  $\pi_3$ .

BC, because there are fewer observations for this earlier sample. The pattern across specifications is the same as in Table 6. Second, I use arable land instead of initial population as the proxy for initial conditions.<sup>14</sup> In this specifications,  $\pi_3$  is also positive in the absence of controls, and negative (though insignificant) once the basic controls are added. I also consider (but do not report) alternative specifications for the Biraben (1979) data. I consider two additional starting dates, 200 BC and 1 AD. Both cases reproduce the patterns seen in Tables 5 and 6. Finally, it may be useful to discriminate across regions given that respective pre-1500 AD developments were independent. I have estimated specification (5) from Table 5 separately for the Old World and the New World. (The fixed effects estimates are representative of all other specifications.) In order to maximize the number of observations, I use  $t_0 = 1$  AD. The estimates are  $\pi_3^{\text{Old World}} = -2.1$  (s.e. 0.3) and  $\pi_3^{\text{New World}} = 0.4$  (s.e. 0.4). For regions in the Old World,  $\pi_3^{\text{Eurasia}} = -2.0$  (s.e. 0.4) and  $\pi_3^{\text{Africa}} = -2.7$  (s.e. 1.0). The only positive but insignificant estimate of the interaction term is for the New World, which also contradicts the idea that areas with larger initial populations have faster population growth.

**Overview** In pre-modern samples, population growth and initial population size are positively associated but only in the absence of control variables that proxy for systematic differences across countries. Indeed, systematic controls fully reverse this association. These reversals, as with previous predictions, suggest an instability in the relationship between population growth and initial population size. This instability is not a consequence of the modern demographic transition, as prediction (iv) has been tested with pre-modern data. The nature of the instability discussed here is related to the possibility of confounding influences in the analysis of the melting of the ice caps. There is, for instance, an interesting contrast in terms of prediction (iv) if one were to apply it to modern data. A positive estimate of  $\pi_3$ suggests momentum dynamics consistent with Boserupian effects. A negative estimate of  $\pi_3$  suggests *mean-reverting dynamics* consistent with Malthusian effects. In samples that include modern population data, populations do not exhibit mean reversion and therefore they are inconsistent with Malthusian dynamics.<sup>15</sup> The negative estimates presented in Tables 5 and 6, however, are consistent with the tendency of populations to return to some stationary and self-correcting Malthusian equilibrium.

To eliminate confounding influences in tests of prediction (iv), one should compare more similar geographic areas. One can, for example, compare tropical areas in the Old and New Worlds since they share some similarities in "exogenous"

<sup>&</sup>lt;sup>14</sup>Population in 200 BC may be an inadequate proxy for the population size prior to the melting of the ice caps. In 200 BC, the large centers of agricultural production in Asia were consolidated and may have faced relatively stagnant conditions. Arable land may be a better proxy for the size of pre-treatment populations.

<sup>&</sup>lt;sup>15</sup>For example, regressing  $\ln[N_{i,t}]$  on  $\ln[N_{i,t-1}]$  in the full sample for McEvedy and Jones (1985) yields a fixed effects point estimate of 1.030 (s.e. 0.016). Since this point estimate exceeds one, the estimates suggest "divergence" across regions. For the reasons discussed in the previous section, this divergence appears to be a transitory event associated with the demographic transition.

	OLS	OLS					
	(1)	(2)	(3)	(4)	(5)		
Data: McEvedy and Jor	nes (1985)						
A. Different initial date	$, t_0 = 1 \text{ AD}$						
$\ln[N_{i,0}] \times \text{Time}$	6.08***	6.14***	- 1.59***	-1.57***	-1.59***		
	(0.27)	(0.25)	(0.35)	(0.32)	(0.30)		
	[0.17]	[0.22]	[0.40]	[0.42]	[0.43]		
Time	No	Yes	Yes	Yes	Yes		
$\ln[N_{i,0}]$	No	No	Yes	Yes	-		
Continental controls	No	No	No	Yes	-		
Obs/Countries	378	378	378	378	378/64		
B. Using arable land ins	stead of initial po	pulation size					
$\ln[Land_i] \times Time$	1.80***	2.36***	-3.13	-3.42	-0.79		
	(0.43)	(0.48)	(9.52)	(7.92)	(2.78)		
	[1.15]	[1.34]	[5.27]	[0.42]	[5.97]		
Time	No	Yes	Yes	Yes	Yes		
$\ln[N_{i,0}]$	No	No	Yes	Yes	-		
Continental controls	No	No	No	Yes	-		
Obs/Countries	416	416	416	416	416/71		

 Table 6
 Population change and initial population before 1500

The dependent variable is log population size. Robust standard errors are in parentheses. In brackets are standard errors clustered at the continental level. The specifications are the same as in Table 5. \*\*\*, \*\*, and \*denote significance at the 1, 5, and 10 percent levels.

conditions and thus are more valid to compare. Their similarities, however, also extended to economic conditions in 1500 AD making definite assessments about the influence of initial population sizes difficult. (I perform a more detailed comparison along the lines suggested here in an Appendix not for publication.) Future work might also explore a different natural experiment. Holland (2000) considered the fate of European cities if Ogadai Khan had not died on the eve of the Mongol siege of Vienna in 1242 AD and estimated the impact using Bagdad as a "control". (After destroying the Christian armies of Poland and Hungary, the Mongols were poised to siege Vienna when Ogadai Khan's death prompted Batu Khan to return to Karakoram to elect a new Khan. Depopulation was fortuitously avoided as the Mongols never returned and Europe.) According to Holland (2000, p. 93), if European cities had experienced Bagdad's massive destruction following the Mongol siege, Europe would have replaced learning with religious prejudice leading to the fundamentalism that the Islamic world experienced after the Mongol invasions; "[t]he Dark Ages were pure light compared to what could have happened to Europe if, in the thirteenth century, it had been overrun by the Mongols".

## 5 An alternative formulation

This section briefly discusses a simple alternative to the previous analytical framework. I first outline some amendments to the baseline theory of Section 2 that allow for transitional patterns. Later in this section, I provide some remarks about the particular assumptions used here.

**Limits to knowledge** Technological leapfrogging and transitional growth seem critical to account for the present empirical findings. In particular, past technological sophistication and population size reinforce technological change in ideas-based models such as (1): the more accumulated experience with a given mode of production, the higher the likelihood of additional innovations. This predicted continuity in technological conditions is inconsistent with the main findings of this paper. Moreover, the knowledge production function (1) is not equipped to deal with transitional population growth.

Consider a modified version of the knowledge production function (1) consistent with technological leapfrogging and transitional growth. The following alternative formulation of the knowledge production function assumes that, within a particular mode of production, there is a (possibly unbounded) frontier of knowledge  $A_+$ . For  $A_+ < \infty$ , as the stock of knowledge increases beyond a certain point, marginal knowledge productivity will decline and become negative. This formulation will therefore be consistent with a "fishing out" externality associated with a finite number of fish to catch; see Jones (2001). There will also be different modes of production according to the value of the frontier  $A_+$ . In particular, as in leapfrogging models; see, e.g., Brezis et al. (1993), technological change will be of two general forms, "normal" and "radical", with radical changes influencing the frontier of knowledge  $A_+$ .

Suppose that the knowledge production follows a logistic growth function given by  $\dot{A}(t) = \lambda A(t)^{\phi} N(t)^{\gamma} [1 - (A(t)/A_{+})^{\kappa}]$ , where  $\kappa$  is a normalizing parameter specified in the context of population growth. Under a "subsistence level"  $\bar{y}$ , population growth also follows a logistic growth function given by

$$\dot{N}(t) = \theta N(t)^{\alpha} \left[ 1 - \left( \frac{N(t)}{N_+} \right)^{\kappa(1-\eta)} \right], \tag{10}$$

where  $N_+ \equiv (A_+/\bar{y})^{1/(1-\eta)}T$ . If  $\kappa = (1-\alpha)/(1-\eta)$ , expression (10) is a Bernoulli differential equation with solution

$$N^{*}(t|\theta, N_{0}, N_{+}) = \left[N_{0}^{1-\alpha} \exp\left\{-\frac{\theta(1-\alpha)}{N_{+}}\right\} + \theta(1-\alpha)\int_{0}^{t} \exp\left\{\frac{\theta(1-\alpha)}{N_{+}}[s-t]\right\} ds\right]^{1/(1-\alpha)},$$

which equals (4) as a special case when population size is unbounded, i.e., when  $N_+ \rightarrow \infty$ .

To understand the implications of Eq. 10, consider first the time series properties of a global population whose initial value is  $N_0$ . First, notice that a bounded value of  $A_+$  implies that population size will be bounded by  $N_+$ . The growth rate of population will also be bounded. In fact, population will follow the traditional S-shaped trajectory with an accelerating segment from  $N_0$  to  $\hat{N} \equiv N_+(1-\alpha^{-1})$  reaching a peak population growth, whose value is  $\theta \hat{N}^{\alpha} / \alpha$ , followed by a decelerating phase from  $\hat{N}$ to  $N_+$ . The early transitional trajectory is especially interesting because this phase is indistinguishable from the baseline Malthus-Boserup model outlined in Section 2. During the initial phase of the transition, for example, population growth and the size of population will be positively related, as prediction (i) suggests. (Predictions (ii)–(iv) also hold in this phase.) Expression (10) predicts, however, that population growth will, at some point, start to slow down and continue slowing down until the transition is complete. In demographic data, the accelerating phase of global population growth that ended in 1960 is easily observable in Fig. 1 and Table 1. Figure 1 also hints of a slow down in population growth consistent with an end to the demographic transition and the decelerating phase predicted by Eq. 10.

Consider also the cross-sectional implications of a diffusion of "radical" technological changes. Suppose that a new technology that increases the frontier of knowledge is introduced. This technology has a higher value for  $A_+$  so it dominates old technologies by increasing the population and economic growth potential. Suppose, as in Brezis et al. (1993), that the adoption of a radical technology is based on comparative advantage. The early adopters of this new technology will be the backward countries because the new technologies are more advantageous at low levels of population and sophistication. As in leapfrogging models, the introduction and adoption of radical technological changes will produce reversals such as the ones documented here; backward economies will slowly overtake the population size and technological sophistication of advanced regions. Such overtaking is consistent with the reversals in the role of exogenous factors documented in Table 3.

**Some remarks** In expression (10), Boserupian effects are present during "normal" times but they are limited to the early stages of a transition, unless the frontier is assumed increasing in population. Since  $A_+$  is given in Eq. 10, a dependence between  $A_+$  and N would represent an additional Boserupian channel, not tested here. The few "radical" technological changes in the present sample (and in history, more generally) make such specification difficult to test.

The absence of Boserupian effects in "radical" technological changes means that an Industrial Revolution is not inevitable. This is a contrast to Jones (2001) and expression (1). The distinction between "normal" and "radical" technological change is also consistent with the persistence in technological sophistication between 1000 BC, 0 AD, and 1500 AD, documented by Comin et al. (2010, Tables 7A and 7B), and with the absence of any significant effect of technological sophistication during 1000 BC and 0 AD on current economic and technological conditions; see, e.g., Comin et al. (2010, Table 8A). Persistence is natural during "normal" times because knowledge within any one particular mode of production is cumulative. Technological leadership, however, is irrelevant during "radical" technological changes. The irrelevance of past technological conditions is notable in Europe's industrialization. Diamond (1997, pp. 409-410), echoing many other social scientists, noted

"A historian who had lived at anytime between 8500 B.C. and 1450 A.D., and who had tried then to predict future historical trajectories, would surely

have labeled Europe's eventual dominance as the least likely outcome, because Europe was the most backward of those three Old World regions for most of those 10,000 years. [...] Until the proliferation of water mills after about A.D. 900, Europe west or north of the Alps contributed nothing of significance to Old World technology or civilization".

Expression (10) has a "limits to growth" flavor but it does not carry an alarmist or fatalist message as there is no presumption that population is near such limit or that  $N_+$  is unmovable. It is not possible to discuss in detail the role of each of the determinants of  $N_+ \equiv (A_+/\bar{y})^{1/(1-\eta)}T$ , but note that physical barriers are the most commonly invoked limit to population growth. Complains about the pressure of population on arable land, central in Malthusian theory, date back to the Bible, and Quintus Septimus Florence Tertillianus, if not earlier; see Johnson (2000, p. 1). Limits to scientific knowledge, on the other hand, are rarely discussed although scientists have identified several limits; see, e.g., Casti and Karlqvist (1996) and Hut et al. (1998). These limits are logical (not practical) and they represent conceptual (not technological) limits of science, not limits of scientists. As Casti and Karlqvist (1996, p. 12) noted, citing several 'impossibility theorems' across scientific disciplines,

"[T]o anyone infected with the idea that the human mind is unlimited in its capacity to answer questions about natural and human affairs, a tour of 20thcentury science must be quite a depressing experience. Many of the deepest and most well-chronicled results of science in this century have been statements about what *cannot* be done and what *cannot* be known".

Finally, population size is limited by subsistence conditions in the form of  $\bar{y}$ . Subsistence, as a general demographic concept, has no place in the modern era for developed countries. Instead, it is possible to abuse the interpretation of the model and argue that parental investments in health and education require an acceptable standard that represents, in a loose way, a subsistence requirement during the modern era. At the present, space and knowledge do not seem binding limits to population size. Behavioral factors that determine parental investments seem to critically limit modern population growth.

### 6 Some final remarks

This paper introduced several new population growth predictions from an ideas-based model of long-term technological change; a model that generalizes Kremer's (1993) and Klasen and Nestmann's (2006) Malthus-Boserup syntheses. It also tested these predictions using numerous alternative data sources, empirical specifications, and sample periods.

In general, and especially outside of the modern era, I found limited support for the hypothesis that population growth spurs technological change. I found that the model's implied relationship between population growth and technological change is very sensitive to the sample period and specification. The sign and statistical significance of the relationships between population growth and size, and between population growth and exogenous factors such as biogeography and arable land, are completely reversed between modern and pre-modern samples. The model also predicts a positive association between population growth and initial population size in isolated populations. The melting of the ice caps provided a *natural experiment* for these tests. In pre-modern samples, however, the association between population growth and initial population size is very sensitive to the controls used; the outcomes of these tests, for example, are typically reversed when confounding influences are taken into account.

It is not possible to conclude unequivocally in favor of a Boserupian view of the long-term relationship between population growth and technological change. The findings, in fact, suggest a sharp contrast between pre-modern and modern technological and demographic regimes. Pre-modern population data exhibit mean reversion consistent with the notion of a stable stationary Malthusian equilibrium; see also Ashraf and Galor (2011). A Mathusian rear-view mirror, however, is not very useful for forecasting future population paths. Although continuity is important in the long-term analysis of population and technology, the main lesson of this paper is that the relationship between population growth and technological change has undergone a major change in the modern era.

There are obviously many statistical caveats in the interpretation and analysis of long-term population data. The data sources are not ideal and segmentation of the data will likely exacerbate measurement biases. Confirmation bias is also important because none of the sources here rely on direct measurements of population size. As discussed by Kremer (1993, pp. 699-700), the basic assumption behind the existing estimates of past population sizes appears to be a Malthusian assumption that associates an increase in population with exogenous technological changes. The statistical analyses also omit many social, economic, and political influences that cannot be measured given the scope of the paper. All these considerations and omissions are important. It is quite unlikely, however, that one can overcome the data limitations in the short term.<sup>16</sup>

There are also many economic caveats in the interpretation offered here. Particularly, the historical record suggests that earlier periods are characterized by small and infrequent discrete technological changes with a slow diffusion across connected world regions. The current tests only rely on "recent" data. Thus, the tests do not necessarily discriminate against very gradual technological change, which perhaps was the norm prior to the Industrial Revolution. Boserupian effects may be harder to detect given the shorter span of data considered here, or it might be that settled agriculture introduced a Malthusian mechanism into an otherwise Boserupian process. The difficulty is that without reliable data it is nearly impossible to examine the presence of Boserupian effects in early human demography or in "radical" technological changes. While this paper does not support extreme pessimism, one can

<sup>&</sup>lt;sup>16</sup>There are few alternative data sources for examining pre-modern conditions. Anthropological analyses of genetic diversity in current populations have been able to shed light on the demography of past populations; see, e.g., Relethford (2001, 2003). In economics, a growing literature has started using genetic information to examine current and past differences in economic development; see, e.g., Spolaore and Wacziarg (2009) and Ashraf and Galor (2013).

read the present findings as a call for caution in interpreting the existing empirical evidence using extreme optimism. The findings also show the need for further theoretical and empirical investigations of the long-term relationship between population and development.

**Acknowledgments** I would like to thank two anonymous referees and the editor for many helpful suggestions. I would also like to thank Gary Becker, Michele Boldrin, David Carr, Robert Fogel, Kellie Forrester, Michael Gurven, Hyeok Jeong, Stephan Klasen, Gary Libecap, Peter Lindert, Bonnie Queen, Romain Wacziarg, Bruce Wyndick, as well as seminar participants from several different locations for their comments. I would especially like thank the late D. Gale Johnson for his many useful discussions on the role of population in the economy.

## References

- Acemoglu D, Johnson S, Robinson JA (2002) Reversal of fortune: geography and institutions in the making of the modern world income distributions. Q J Econ 107:1231–1294
- Acemoglu D, Aghion P, Bursztyn L, Hemous D (2012) The environment and directed technical change. Am Econ Rev 102:131–166
- Ang JB, Madsen JB (2011) Can second-generation endogenous growth models explain the productivity trends in the Asian miracle economies? Rev Econ Stat 93:1360–1373

Ashraf Q, Galor O (2011) Dynamics and stagnation in the malthusian epoch. Am Econ Rev 101:2003– 2041

Ashraf Q, Galor O (2013) Human genetic diversity and comparative economic development. Am Econ Rev 103:1–46

Bairoch P (1988) Cities and economic development: from the dawn of history to the present. University of Chicago Press

- Baland J-M, Robinson JA (2002) Rotten parents. J Public Econ 84:341-356
- Biraben J-N. (1979) Essai sur L'é volution du Nombre des Hommes. Population 34:13-25
- Bohn H, Stuart C (2015) Calculation of a population externality. Am Econ J Econ Policy 7:61-87

Boserup E (1981) Population and technology. Wiley-Blackwell

- Brezis ES, Krugman PR, Tsiddon D (1993) Leapfrogging in international competition: a theory of cycles in national technological leadership. Am Econ Rev 83:1211–1219
- Caldwell JC, Schindlmayr T (2002) Historical population estimates: unraveling the consensus. Popul Dev Rev 28:183–204
- Casti JL, Karlqvist A (1996) Boundaries and barriers: on the limits to scientific knowledge. Addison-Wesley
- Chatterjee S, Hadi AS (1988) Sensitivity analysis in linear regression. Wiley
- Cigno A (1981) Growth with exhaustible resources and endogenous population. Rev Econ Stud 48(2):281–287
- Clark C (1967) Population growth and land use. Macmillan and St. Martin's Press
- Cohen JE (1995) Population growth and earth's human carrying capacity. Science 269(5222):341–346
- Collard M, Buchanan B, O'Brien MJ, Scholnick J (2013) Risk, mobility or population size? Drivers of technological richness among contact-period Western North American hunter-gatherers. Philos Trans R Soc B 368:2012.0412
- Comin D, Easterly W, Gong E (2010) Was the wealth of nations determined in 1000 BC? Am Econ J Macroecon 2:65–97
- Dasgupta P (2000) Population and resources: an exploration of reproductive and environmental externalities. Popul Dev Rev 26:643–689
- Deevey E (1960) The human population. Sci Am 203(3):194-204
- Deevey E (1968) Pleistocene family planning. In: Lee RB, DeVore I (eds) Man the hunter. Aldine Publishing Company
- Diamond J (1997) Guns, germs and steel: the fates of human societies, Vintage
- Fagan B (2005) The long summe: how climate changed civilization. Grata Books
- von Foerster H, Mora P, Amiot L (1960) Doomsday: Friday 13, November A.D. 2026. Science 132(3436):1291–1295

Galor O, Weil D (2000) Population, technology and growth: from malthusian stagnation to the demographic transition and beyond. Am Econ Rev 90:806–828

Harlan J (1992) Crops and man. American Society of Agronomy

Hassan FA (1981) Demographic archaeology. Academic Press

Headey DD, Hodge A (2009) The effect of population growth on economic growth: a meta-regression analysis of the macroeconomic literature. Popul Dev Rev 35:221–248

Henrich J (2004) Demography and cultural evolution: how adaptive cultural processes can produce maladaptive losses: the tasmanian case. Am Antiq 69:197–214

Hibbs DA Jr, Olsson O (2004) Geography, biogeography, and why some countries are rich and others are poor. Proc Natl Acad Sci 101:3715–3720

Holland CA (2000) The death that saved europe: the mongols turn back. In: Cowley R (ed) What if?: the world's foremost military historians imagine what might have been. GP Putman's Sons, pp 93–106

Hut P, Ruelle D, Traub J (1998) Varieties of limits to scientific knowledge. Complexity 3:33-38

Johnson DG (2000) Population, food, and knowledge. Am Econ Rev 90:1-14

Jones CI (2001) Was an industrial revolution inevitable? Economic growth over the very long run. Adv Macroecon 1:1–43

Jones CI (2005) Growth and ideas. In: Aghion P, Durlauf S (eds) Handbook of economic growth, vol 1. Elsevier

Kelley AC (1988) Economic consequences of population change in the third world. J Econ Lit 26:685– 1728

Klasen S, Nestmann T (2006) Population, population density and technological change. J Popul Econ 19:611–626

- Kline MA, Boyd R (2010) Population size predicts technological complexity in oceania. Proc R Soc B 277:2559–2564
- Kortum S (1997) Research, patenting, and technological change. Econometrica 65:1389-1419
- Kremer M (1993) Population growth and technological change: one million B.C. to 1990. Q J Econ 88:681–716
- Kuznets S (1960) Population change and aggregate output. In: Demographic and economic change in developed countries. Princeton University Press
- Lee RD (1988) Induced population growth and induced technological progress: their interaction in the acceleration stage. Math Popul Stud 1:265–288
- Lee RD, Miller T (1990) Population growth, externalities to childbearing, and fertility policy in developing countries. In: Proceedings of the World Bank Annual Conference on Development Economics, World Bank
- Maddison A (2001) The world economy: a millennial perspective, development centre of the organisation for economic co-operation and development
- Madsen JB (2008) Semi-endogenous versus Schumpeterian growth models: testing the knowledge production function using international data. J Econ Growth 13:1–26
- Madsen JB, Ang JB, Banerjee R (2010) Four centuries of British economic growth: the roles of technology and population. J Econ Growth 15:263–290

McEvedy C, Jones R (1985) Atlas of world population history. Penguin Books

- Pryor FL, Maurer SB (1982) On induced economic change in precapitalist societies. J Dev Econ 10:325– 353
- Pritchett L (1996) Population growth, factor accumulation, and productivity. World Bank Policy Research Paper 1567
- Ravallion M (2010) Population scale effects revisited. World Bank Working Paper
- Read D (2006) Tasmanian knowledge and skill: maladaptive imitation or adequate technology? Am Antiq 71:164–184
- Read D (2012) Population size does not predict artifact complexity: analysis of data from Tasmania, Arctic hunter-gatherers, and Oceania Fishing Groups, UCLA Mimeo
- Relethford J. H (2001) Genetics and the search for modern human origins. Wiley-Liss Press
- Relethford JH (2003) Reflections of our past: how human history is revealed in our genes. Westview Press

Segerstrom PS (1998) Endogenous growth without scale effects. Am Econ Rev 88:1290-1310

- Simon J (1977) The economics of population growth. Princeton university press
- Smith BD (1995) The emergence of agriculture. Scientific American Library

Spolaore E, Wacziarg R (2009) The diffusion of development. Q J Econ 124:469–529

Umpleby A (1987) World population: still ahead of schedule. Science 237(4822):1555-1556

Whitmore TM, Turner BL II, Johnson DL, Kates RW, Gottschang T. R. (1990) Long-term population change. In: The earth as transformed by human action. Cambridge University Press